Author's response to reviews

Title: Burden of frailty in the elderly population: perspectives for a public health challenge

Authors:

Fanny BUCKINX (fanny.buckinx@ulg.ac.be)
Yves ROLLAND (rolland.y@chu-toulouse.fr)
Jean-Yves Reginster (jyrejinster@ulg.ac.be)
Celine RICOUR (celine.ricour@chu.ulg.ac.be)
Jean PETERMANS (Jean.petermans@chu.ulg.ac.be)
Olivier BRUYERE (olivierBruyere@ulg.ac.be)

Version: 4
Date: 6 February 2015

Author's response to reviews: see over
Editor's comment
Title: Burden of frailty in the elderly population: perspectives for a public health challenge
Authors:
Fanny BUCKINX (fanny.buckinx@ulg.ac.be)
Yves ROLLAND (rolland.y@chu-toulouse.fr)
Jean-Yves REGINSTER (jyreginster@ulg.ac.be)
Celine RICOUR (celine.ricour@chu.ulg.ac.be)
Jean PETERMANS (Jean.petermans@chu.ulg.ac.be)
Olivier BRUYERE (olivier.bruyere@ulg.ac.be)
Version: 3
Date: 6 February 2015
Editor's comment:
Dear authors,
many thanks for providing the additional information summarized in figure 1. May I ask you to (1) add a title to the figure (the title should be sufficient specific so that the figure can be understood without the text)) and (2) to review the journal guidelines related to the submission requirements for figures.

Author’s response:

Dear Editor,

As requested, the title and the format of the figure have been revised.